Author’s response to reviews

Title: Anti-Hu Antibody associated Paraneoplastic Cerebellar Degeneration in Head and Neck Cancer

Authors:

Florian Huemer (florian.huemer@pmu.ac.at)
Thomas Melchardt (t.melchardt@salk.at)
Wolfgang Tränkenschuh (w.traenkenschuh@salk.at)
Daniel Neureiter (d.neureiter@salk.at)
Gerhard Moser (g.moser@salk.at)
Teresa Magnes (t.magnitude@salk.at)
Lukas Weiss (lu.weiss@salk.at)
Alexander Schlattau (a.schlattau@salk.at)
Clemens Hufnagl (cl.hufnagl@salk.at)
Gerda Ricken (gerda.ricken@meduniwien.ac.at)
Romana Höftberger (romana.hoelfberger@meduniwien.ac.at)
Richard Greil (r.greil@salk.at)
Alexander Egle (a.egle@salk.at)

Version: 4
Date: 10 December 2015

Author’s response to reviews:

As requested, distinction between the authors Thomas Melchardt and Teresa Magnes has been carried out in the authors’ contributions section.